



# Article ASIPAMPIUM: An Efficient ASIP Generator for Low Power Applications

Alian Engroff, Marcelo Romanssini \*, Lucas Compassi-Severo \*<sup>D</sup>, Paulo C. C. de Aguirre \*<sup>D</sup> and Alessandro Girardi \*<sup>D</sup>

Alegrete Campus, Federal University of Pampa, Av. Tiaraju 810, Alegrete 97564-550, RS, Brazil

\* Correspondence: marceloromanssini.aluno@unipampa.edu.br (M.R.); lucas.severo@unipampa.edu.br (L.C.-S.); paulo.aguirre@unipampa.edu.br (P.C.C.d.A.); alessandro.girardi@unipampa.edu.br (A.G.)

Abstract: The adoption of customized ASIPs (Application Specific Instruction Set Processors) in embedded circuits is an important alternative for optimizing power consumption, silicon area, or processing performance according to the design requirements. The processor is implemented specifically for the target application, which allows the hardware customization in terms of instruction set architecture, data word length, memory size, and parallelism. This work describes an EDA tool for the semi-automatic development of ASIPs named ASIPAMPIUM. The strategy is to provide a set of integrated tools to interpret and generate a customized hardware for a given target application, including compilation, simulation, and hardware synthesis. From the C code description of the application, the tool returns a synthesizable hardware description of the processor. The proposed methodology is based on the adaptation of a new customizable microprocessor called PAMPIUM, which can be optimized in terms of silicon area, power consumption, or processing performance according to the target application. The ASIPAMPIUM tool provides a series of simulated data to the designer in order to identify optimization strategies in both software and hardware domains. We show the results for the implementation of an FFT algorithm using the proposed methodology, which achieved best results in terms of silicon area and energy consumption compared to other works described in the literature for both FPGA and silicon implementation. Moreover, measurement results of the implementation in silicon of a dedicated ASIP for interfacing with six sensors in real-time, including three I2C, an SPI, and an RS-232 interfaces, demonstrate the complete design flow, from the C code program to physical implementation and characterization. Aside from providing a short design time, the ASIPAMPIUM tool also affords a simple and intuitive design flow, allowing the designer to deal with different design trade-offs and objectives.

Keywords: ASIP; EDA tool; FFT

# 1. Introduction

In the last decades, there has been great progress in the development of applicationspecific embedded systems. The advances in micro- and nanoelectronic device fabrication technology and the development of new design techniques, which allow the implementation of systems-on-chip (SOCs), are key for the implementation of mobile applications and Internet of Things (IoT). These applications have the strong requirement of very low energy consumption, often combined with high performance [1].

The most important element of an embedded system is the microprocessor. This element must be, in some cases, developed specifically for a given system, which has requirements in terms of circuit area, power consumption, and performance, and is dependent on the application and on the implemented functionalities [2]. With the growing demand caused by IoT, new functionalities are required, which increases design complexity [3,4]. Furthermore, there is strong pressure for fast implementation of technological solutions in the globalized market. This forces companies to design new embedded systems



Citation: Engroff, A.; Romanssini, M.; Compassi-Severo, L.; de Aguirre, P.C.C.; Girardi, A. ASIPAMPIUM: An Efficient ASIP Generator for Low Power Applications. *Electronics* 2023, 12, 401. https://doi.org/10.3390/ electronics12020401

Academic Editors: Chiper Doru Florin and Constantin Paleologu

Received: 9 December 2022 Revised: 3 January 2023 Accepted: 7 January 2023 Published: 12 January 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). within ever narrower time windows. In addition, new products have a shorter service life, so that the financial return must be obtained in a very short time [5].

In this context, the development of ASIPs (Application-Specific Instruction Set Processors) is a good alternative for fast and optimized system implementation. The customizable processors are designed for a given application domain, with the instruction set specifically designed to accelerate heavy and most frequent functions. The architecture must implement the assembly instruction set with minimum hardware cost. The main advantages are flexibility and reduced time to market [6].

Custom processors are often needed for optimizing application characteristics, mainly performance under a strict power consumption budget. This leads to the development of custom platforms to support design stages and application evolution. ASIP design methodologies have stagnated during the past decade and are still based on a series of manual and time-consuming iterative steps. Furthermore, a productivity gap exists between the point where an application is given as the target for processor customization and the time a customized architecture is available. The flexibility and reduced time to market are important advantages of using ASIPs. Therefore, new EDA tools are required for reducing the number of design iterations and to bridge this productivity gap [7].

In most applications, the ASIP technology requires performance and energy usage compared to those of hardwired ASICs [8]. Some challenges arise in this field when stringent and conflicting specifications demand highly-optimized hardware architectures, which requires the exploration of hardware solutions with different design characteristics. The adoption of parallelism is, in most cases, an important strategy to improve performance, but increases design complexity in terms of both hardware and embedded software.

The presence of development platforms for the design of customized processors is mandatory for fast system development. Furthermore, it must allow the possibility to perform design evolution and reusability since the designed system must continuously follow the technology trends. The efficient mapping of complex applications by these platforms is a hard requirement. The EDA tool must have an effective uniform synthesis flow for HW/SW compilation of high-level behavioral specifications. This compilation process should include an adequate application-specific reconfigurable hardware platform synthesis and a corresponding compilation of high-level algorithms into their highly optimized code executed on the synthesized application-specific hardware platform [9].

Customized applications must achieve a series of requirements in order to deal with environmental and performance conditions. It must guarantee real-time performance, high safety, security, and dependability, while at the same time satisfying the requirement of low energy consumption. A specialized EDA tool for this task must deal with complex multi-level multi-objective optimization (e.g., performance versus energy consumption and silicon area), besides attending to complex design trade-offs originated by conflicting requirements [1]. At the same time, the EDA tool must increase design productivity, even for complex and sophisticated systems, and reduce time-to-market and development costs without compromising system quality.

The ideal system development flow would be a seamless compilation of a high-level computation process specification (e.g., in C code) into an optimized machine code executed on customizable processors [9]. The performance of CISC and RISC architectures can be improved by exploring instruction level parallelism (ILP), including strategies such as pipeline, superscalar, and very long instruction word (VLIW).

The capability to deal with conflicting design requirements according to the targeting application—such as speed, energy consumption, and circuit area—while accelerating the ASIP development would be a valuable contribution to this field. In this sense, we present a new EDA tool for the semi-automatic development of ASIPs named ASIPAMPIUM. The strategy is to provide a set of tools and methods to interpret and generate customized hardware for a given target application, including compilation, simulation, and hardware synthesis. It is based on the adaptation of a new customizable microprocessor called PAMPIUM, which has three baseline micro-architectural versions: monocycle, pipeline,

and superscalar. The monocycle version is used for simple applications, in which small circuit area and low energy consumption are priorities. The pipeline version is characterized by the parallel execution at instruction level, increasing the throughput and the performance of the generated hardware. The superscalar version provides parallelism of arithmetic operations and data access, increasing the performance at the expense of an increase in area and power consumption. The advantage of using PAMPIUM architecture is that its simple architecture allows the complete characterization of the hardware features, providing the designer with a rich view of design options for dealing with specification trade-offs.

The PAMPIUM microprocessor has the necessary flexibility to adapt to several applications with different characteristics, allowing the use of a single development platform for different design objectives. The integration of these functionalities into a single EDA tool minimizes the rapidly growing design productivity gap, thus reducing development costs. The ASIPAMPIUM tool provides these features and allows the efficient exploration of design space in the search for an optimized ASIP solution.

The remainder of this paper is organized as follows: Section 2 presents a review of related works in the literature; Section 3 introduces the architecture and organization of the proposed microcontroller called PAMPIUM; Section 4 describes the ASIPAMPIUM tool; Section 5 shows an example of application of the ASIPAMPIUM tool for the fast design of an FFT module; Section 6 presents the silicon implementation of PAMPIUM and its electrical validation; and finally, Section 7 presents the concluding remarks.

## 2. Related Work

There are several works in the field of automatic development of ASIPs. In [8], the proposed approach concentrates on dominating the automatic synthesis and mapping of a given application to heterogeneous massively-parallel MPSoCs based on ASIPs. The technique is based on ASAM (*Automatic Architecture Synthesis and Application Mapping*). Each ASIP of the platform forms a VLIW machine capable of executing parallel software with a single thread of control. Focused on parallel architectures to improve performance, the goal is to build adaptable ASIP-based MPSoCs at substantially lower costs and with shorter time-to-market than the hardwired ASICs. However, it presents small flexibility to applications where the circuit area and power consumption are the main constraints.

The work in [7,10] proposes a global flow for application mapping called CoEx. The approach is oriented to multigrained profiling (MGP), which identifies the profiling needs at each step of ASIP design and allows the designer to tailor the level of detail for application inspection. Moreover, it tries to optimize the hardware and the application algorithm. A prearchitectural estimation engine performs a preliminary report for an application with an abstract processor model and generates an estimate on of the achievable performance.

The platform proposed by [11] explores application parallelism to fulfill performance requirements focused on digital communication applications. It is structured around configurable ASIPs combined with an efficient memory and communication interconnect scheme. The designed ASIP has a single instruction multiple data (SIMD) architecture with a specialized and extensible instruction-set and six-stage pipeline control.

A tool for the development of ASIPs based on the 32-bit LEON3 processor is presented in [12]. The tool analyses the application in order to optimize the circuit area while keeping the performance. The focus is on parallel low-complexity applications.

The work presented by [13] uses LISA 2.0 to generate the hardware. The main goal of LISA is to detect the details of the processor architecture implementation and to model it to a higher level of abstraction. It is based on the VLIW approach with high levels of parallelism. LISA provides an architecture description language and automatically creates both the hardware description and software support tools. The LISA language provides high flexibility to describe the instruction-sets of various processors, such as SIMD, MIMD, and VLIW-type architectures [14].

The RISC-V [15] is an open-source configurable architecture that provides a programmable processor base for custom accelerators. An open-source System-on-Chip design generator tool called Rocket Chip [16] is capable of automatically generating synthesizable RTL. It leverages the Chisel hardware construction language to compose a library of so-phisticated generators for cores, caches, and interconnects into an integrated SoC. Another tool, called OpenASIP 2.0 [17], enables RTL generation as well as high-level language programming of RISC-V processors with custom instructions.

Commercial toolsets, such as Synopsys ASIP Designer [18], Codasip Studio [19] and Andes [20], are available to the industry to provide rapid exploration of architectural choices, generation of C/C++ based software development kits, and automatic generation of power and area-optimized synthesizable RTL.

Most of the present works in this field are devoted to a specific application niche. Most of the tools are optimized for applications with high level of parallelism and have the goal to increase the execution performance and to generate an efficient architecture in terms of power consumption. However, they do not deal efficiently with conflicting design requirements, such as high performance, low energy consumption, and small circuit area. In this context, the elaboration of a flexible tool for accelerating the ASIP development while providing the possibility to deal with design trade-offs would be a contribution to this area. This is the goal of the proposed ASIPAMPIUM tool described in this paper.

## 3. PAMPIUM Architecture

PAMPIUM is a fully customizable microprocessor developed specifically for the design of ASIPs. It is a RISC architecture with a full instruction set composed of 80 instructions. The flexibility of the internal organization allows the configurability for use in different applications since the hardware can be optimally customized for a given design requirement.

The basic internal organization contains a program memory, a data memory, an arithmetic logic unit (ALU), a register bank, and a control unit. There is the possibility to include n in-out ports for external communication, according to application requirements. The basic block diagram of this architecture is shown in Figure 1.



Figure 1. Basic organization of the proposed microcontroller PAMPIUM.

It is possible to configure several aspects, such as the number and types of implemented instructions, number of registers in the register bank, data memory size, type of memory access, instruction word length, data word length, etc. Furthermore, the datapath can be customized by altering the internal organization (monocycle, multi-cycle, pipeline or superscalar). It allows the optimization of the overall characteristics of the microprocessor, resulting in power consumption and silicon area reduction, and performance increase.

It is possible to define customization intervals for some characteristics. For example, the size of the data word is defined by the data type or by the maximum value a variable can achieve (if it is known). Thus, the minimum number of bits for representing an integer variable can be given by the maximum value these variables can achieve in the application.

The size of the instruction word can also be defined by the application. It depends on the number of implemented instructions and registers. It ranges from 9 bits for the basic implementation (small possible ISA) to 40 bits for the full instruction set implementation.

Other parameters, such as the number of registers, depend on the number of variables employed in the application. The register type is given according to the data type (integer, float, etc), which, in its turn, defines the implementation of the functional units (e.g., ALU). Memory access is done according to data type, using indirect addressing, by means of a value given in the instruction word and other stored into a specific register.

PAMPIUM instructions can be divided into three subsets: basic, operative, and data conversion. The basic subset is independent of the data type and contains 13 instructions, as shown in Table 1.

Mnemonic	Short Description				
NOP	No operation				
END	Freeze the program				
JUMP	Jump				
CALL	Jump and save return address				
RET	Function return				
RETL	Function return saving literal				
BBCLEAR	Branch if bit equal to "0"				
BBSET	Branch if bit equal to "1"				
SHR	Shift right				
SHL	Shift left				
BSET	Set a bit from a register				
BCLEAR	Clear a bit from a register				
SETP	Set data memory pointer				

Table 1. Basic instruction subset.

The operative subset is presented in Table 2. It is composed of 20 instructions, which can be used for different data types. The "\*" character can be substituted according to the following data types: "F" for operands of type *Float*, "I" for *Int*, "C" for *Char* and "D" for *Double*.

Table 2. Operative instruction subset.

Mnemonic	Short Description
*СОРҮ	Copy values between registers
*MOVL	Assign literal to #*REG_L
*RM	Read data memory
*WM	Write in data memory
*BL	Branch if larger
*BS	Branch if smaller
*BLE	Branch if large or equal
*BSE	Branch if small or equal
*BE	Branch if equal
*BNE	Branch if no equal
*MULT	Multiplication
*DIV	Division
*ADD	Sum
*SUB	Subtraction
*COMP	Complement
REM	Division remainder
OR	Logic OR
AND	Logic AND
XOR	Logic XOR
NOT	Logic NOT

The instructions in the data conversion subset convert data from one type to another, according to presented in Table 3.

Mnemonic	Short Description
ITOF	Convert integer to float
ITOD	Convert integer to double
FTOI	Convert float to integer
DTOI	Convert double to integer
FTOD	Convert float to double
DTOF	Convert double to float

Table 3. Data conversion instruction subset.

From the 80 different instructions available in the PAMPIUM standard ISA, the user can choose to implement only those useful for the target application.

The instructions are represented in three different formats: *R*, *L*, and *M*. The instruction word is divided into two to four fields, according to the format, as can be seen in Figure 2. The OPCODE field is common to all formats and it stores a unique identification for the instruction. RA, RB, and RC indicate register addresses in the register bank. Fields Literal and INC store constants. The sizes of all fields are configurable, according to the number of implemented instructions and registers.



Figure 2. PAMPIUM instruction formats. (a) R Format. (b) L Format. (c) M Format.

#### 3.1. Electrical Characterization and Modeling

The logic blocks that compose PAMPIUM are electrically characterized in order to provide information to the customized microprocessor generation. This characterization includes the estimation of area, power consumption, and delay time. As each block has configurable specifications, it is important to verify the effect of these specifications on the microprocessor overall performance.

Program memory, register bank, data memory, and arithmetic-logic unit were characterized in 0.18  $\mu$ m CMOS process technology using Synopsys Design Compiler tool. The data obtained for each block are the circuit area, static power consumption, energy per operation, and delay time. The characterization was performed for several word lengths, with the goal to relate word size to the overall circuit characteristics.

All blocks were described in System Verilog and synthesized to gate level. A vector with 10,000 random values was applied to the input of the circuits for simulation in order to obtain a wide coverage over the entire operating range. For the generation of this vector, we used a random-number generator uniformly distributed over the interval [0, 1]. Binary

operands were obtained by multiplying these random numbers by  $2^n$ , in which *n* is the word length.

A model extracted from synthesis results for each building block is used for statistical simulation in which it is possible to determine the more appropriate configuration to be used in the customized microprocessor. These models are based on the simulated results for area, power consumption, delay time, and energy per operation at gate level. Each characteristic is modeled in relation to the size of the block (in the case of memory and register bank) and with respect to instruction or data word size. Figure 3 depicts the model extraction procedure. The block model can be easily extracted from the results obtained by the logic synthesis for different configurations. For example, the area occupied by the program memory is estimated based on the number of instructions it contains and on the instruction word length. A set of synthesis points is generated for a given range in order to evaluate the behavior of this characteristic. So, regression analysis is used to fit a predictive model from the observed points, which provides a generic function of the memory silicon area in relation to the instruction length and to the program size. This model is dependent on the target fabrication process, since the logic synthesis requires information about the fabrication technology. The linear model is used when the modeled characteristic approximates the linear behavior, which happens in most of the situations. Otherwise, other types of approximations can be used that best fit, such as logarithm or exponential functions.



Figure 3. Scheme of model extraction procedure for the PAMPIUM basic blocks.

By means of statistical simulation, it is possible to determine the more appropriate configuration of each block to be used in the customized microprocessor. Moreover, the optimum word length can be estimated. The generated microprocessor can have monocycle, pipeline, or superscalar organization, according to the performance estimated by each compounding block and to the design goals defined by the user. The next subsections describe the characterization and modeling of the main PAMPIUM building blocks.

# 3.1.1. Program Memory

The program memory is a read-only memory for storing program instructions. Its size is determined by the instruction word length and by the size of the application program. The instruction word length can vary, depending on the quantity of registers in the register bank and on the number of different instructions implemented in the customized microprocessor.

Figure 4 presents the logic block of the program memory. The length of input *Addr* (address of the current instruction) is proportional to log base two of the number of instructions stored in the program memory. The input *N\_Read* indicates how many instructions should be read simultaneously at the output (available only in the superscalar version). Outputs *INST*, *INST2*, *INST3* and *INST4* have the length equal to the instruction word.



Figure 4. Logic block for the program memory.

Figure 5 shows the estimation of area, delay, static power and energy per operation (instruction reading) in terms of the length of the instruction word (16, 24 and 32 bits) and of the memory size. It is possible to verify that all characteristics with respect to the number of instructions can be roughly approximated by a linear trend line, thus indicating a clear linear behavior, although a small non-linearity appears. It provides an intuitive scenario for the user, which can easily estimate the effect of reducing or increasing the number of instructions in the program memory. The area of the program memory increases with the number of instructions, as expected (Figure 5a). More instructions mean more bits to be stored in the memory matrix, but additional resources such as line and column decoders, as well as routing, are also necessary. The reading latency (delay in Figure 5b) also increases with the number of instructions. It is caused by the additional routing, which increases the parasitic capacitance of bit lines, slowing down the reading procedure. Line and column decoders also become slower with size, contributing to the performance loss of the memory unit. The same occurs to the static power consumption (Figure 5c) and the energy per operation (Figure 5d). An increase in the number of logic elements results in a growing static power consumption and in more energy expended for executing a single memory reading operation.

Similar behavior is observed in the program memory characteristics with respect to the size of the instruction word. They present a well-defined proportional variation for instructions with 16, 24, and 32 bits, with the exception of energy per operation, which is not significantly affected by the number of bits in the instruction word.



**Figure 5.** Program memory characterization in terms of memory size for different instruction lengths (16, 24, and 32 bits). Trend lines (dotted) indicate the linear regression used for modeling the behavior of the characteristics. (a) Area. (b) Delay. (c) Static power. (d) Energy per operation.

# 3.1.2. Register Bank

The number of registers in the register bank is proportional to the number of variables in the application. The length of each register is determined in order to represent the entire range of values that the variables can achieve. Figure 6 presents the logic block of the register bank for monocycle and pipeline versions. It is possible to simultaneously write data in a register (*R1*) and read two other registers (*R2* and *R3*). Input signals *Addr\_R1*, *Addr\_R2* and *Addr\_R3* represent the addresses of these registers, respectively.



Figure 6. Logic block of the register bank for monocycle and pipeline versions.

The control signal *Set\_P* and the output signal *Pointer* are implemented if data memory read and write instructions are implemented in the instruction set. Control signals *W\_LI* and

*W\_LF* are implemented only if there are instructions for writing integer and floating point literals on registers. In this case, these literals appear at inputs *L\_INT* and *L\_FLOAT*, respectively.

The characteristics of the register bank are affected by the number of implemented registers. Thus, the final results for area, power consumption, and delay depend on the number of registers and on the word length of each register. Figure 7 presents the influence of the word length over the register bank characteristics. It is possible to notice the linear relationship of area, power consumption, and energy per operation with respect to data word length. On the other side, the delay is not affected by the number of bits in the word length.



**Figure 7.** Register bank characterization in terms of data word length. (**a**) Area. (**b**) Delay. (**c**) Static power. (**d**) Energy per operation.

# 3.1.3. Floating Point Unit

The implementation of the floating point unit (FPU) depends on the number of bits for representing the mantissa and the exponent of the floating point number. The overall block size is related to the number of implemented floating point arithmetic operations, which, in turn, are determined by the existence of the respective operation in the instruction set of the customized microprocessor. Figure 8 presents the logic block of the FPU, in which *A* and *B* are the input operands, *RES* is the output result and *SEL* is a control signal indicating the operation to be performed (sum, subtraction, multiplication or division).

Signals *EN\_M* and *EN\_D* enable the start of a multicycle multiplication and division operation, respectively, and signal *T* indicates the end of a multicycle operation. They are implemented only in case of the existence of multicycle multipliers or dividers.

The electrical characterization of this block was achieved by simulation with the number of bits of the mantissa varying from 7 to 23 while keeping the exponent size constant with 8 bits. The following arithmetic blocks are available for implementation in the PAMPIUM: floating point adder, floating point multiplier (monocycle and multicycle versions), and floating point divider (monocycle and multicycle versions).



Figure 8. Logic block of the floating point unit.

Figure 9 shows the simulation results at gate level for these blocks. They present a roughly linear growth for area, delay, and static power consumption with respect to the number of bits in the mantissa. The energy per operation exhibits a near logarithm behavior in relation to data word length for all arithmetic blocks, so it is modeled as a logarithm function of mantissa size. For the multicycle versions, the number of cycles to execute an arithmetical operation depends on several factors, for example, the number of bits *1* and the absolute value of the operands. Thus, this estimative was determined by the average of 10,000 randomly selected input operands along the operation range. The resulting average cycles are indicated in Figure 9b and it also presents a linear behavior in relation to the mantissa size.



**Figure 9.** Floating point unit characterization in terms of mantissa size. (**a**) Area. (**b**) Delay. (**c**) Static power. (**d**) Energy per operation.

### 3.1.4. Integer Arithmetic Unit

The integer arithmetic unit includes operations performed between registers of integer type. The word length is determined by the size of the largest register of this type. Figure 10 presents this logic block. Operands *A* and *B* perform the arithmetic operation determined by the control signal *SEL* and the result is available at the output *RES*. Signals *EN\_D*, which

enables the execution of the multicycle division, and *T*, which indicates the end of the operation, are implemented only in case of the use of multicycle division. The arithmetic blocks that compose the integer arithmetic unit are implemented according to the existence of the operation in the customized instruction set.



Figure 10. Logic block of the integer arithmetic unit.

The electrical characterization was performed for operands with a word length from 8 to 32 bits. Figure 11 presents results for area, delay, static power, and energy per operation for the adder/subtractor, multiplier, and divider blocks in relation to the data word length. The divider has monocycle and multicycle versions. The number of cycles for the execution of an operation in the multicycle divider is dependent on the input values, as for the FPU module. So, the characterization of this block is also the result of a mean of a vector of 10,000 randomly selected operands along the operation range. All electrical characteristics present a near-linear increase as the number of bits of the operands increase, so the model is extracted as a linear function of data length. The number of cycles necessary to perform a division also increases linearly in relation to the word length. Simulation results presented an average number of 5, 7, 9, and 11 cycles for word lengths of 8, 16, 24, and 32 bits, respectively.



**Figure 11.** Integer arithmetic unit characterization in terms of data word length. (**a**) Area. (**b**) Delay. (**c**) Static power. (**d**) Energy per operation.

# 3.2. PAMPIUM Monocycle

The monocycle version of PAMPIUM is composed of the following main logic blocks: program memory, program counter, register bank, and control unit. The remaining blocks are added according to the demand of the target application, including data memory, integer and floating point arithmetic units, and external memory interface. Figure 12 depicts a simplified view of the internal organization of the monocycle version of PAMPIUM. The control block was not shown for better visualization. The datapath is very simple. All instructions are executed in a single clock cycle. This microprocessor is indicated for applications with hard area constraints and relaxed performance requirements. The critical path is determined by the existence and size of the logic blocks, which are implemented according to the customized ISA.



Figure 12. Simplified view of the monocycle architecture of PAMPIUM.

# 3.3. PAMPIUM Pipeline

The pipeline version is based on the monocycle version, but with instruction parallelization. The datapath is divided into four stages: instruction fetch, data fetch, execution and storage. Pipeline registers must be added between the stages, as well as a hazard control unit. Figure 13 presents a simplified internal organization of the pipeline version of PAMPIUM.



Figure 13. Simplified view of the pipeline architecture of PAMPIUM.

The main logic blocks present in this version are the same as in the monocycle version. Some multiplexers need to be added to select the correct datapath in case of data hazard. The internal organization of the processor prevents the existence of structural hazards, because each instruction step is executed at a different stage with independent modules, and the execution automatically freezes in case of data dependency.

Interruptions in the pipeline flow can also occur caused by conditional branches, which are calculated in the second pipeline stage (data fetch).

The presence of pipeline registers represents a huge impact on the microprocessor characteristics. The bigger the data word length, the bigger the effect of registers over circuit area and on power consumption. On the other side, the pipeline organization provides advantages in processing performance, increasing data throughput. This type of organization is indicated for applications that require an equilibrium between power consumption and performance.

#### 3.4. PAMPIUM Superscalar

The superscalar version of PAMPIUM presents higher level of parallelism than the pipeline version, including datapath and instruction level parallelism. It presents eight different datapaths, providing the possibility to execute in parallel the following operations: one integer instruction, the most used integer instruction, one floating point instruction, the most used floating point instruction, one integer literal instruction, one floating point literal instruction, one memory access instruction and one pointer definition instruction. Figure 14 presents the hardware organization for the PAMPIUM superscalar version.

The implemented version can execute up to four instructions per clock cycle. For this, the Program Memory must read up to four instructions simultaneously. The datapath is complex in order to deal with the high number of data hazard possibilities, demanding more pipeline registers, which increases circuit area. However, the parallel execution increases the energetic density, making this version suitable for high performance applications.



Figure 14. Simplified view of the superscalar architecture of PAMPIUM.

## 4. ASIPAMPIUM Tool

The EDA tool for semi-automatic ASIP design proposed in this work, called ASI-PAMPIUM, receives as input an application code in C language and performs analysis and simulation in order to guide the user to the generation of an optimized microprocessor tailored for that application. Circuit area, power consumption, processing time, or a combination of these, can be set as the design objective.



The design flow is shown in Figure 15. It can be divided into four phases [21].

Figure 15. Design flow of the ASIPAMPIUM tool.

In Phase 1 the application code is compiled in order to identify the necessary hardware operations. The main functional units are identified based on software requirements, including adder, multiplier, divider, memory access, etc. Software variables are mapped to memory positions after their types, range of values, and vector sizes are estimated by simulation. With this procedure, it is possible to optimize memory units and data word length.

In Phase 2 the overall design strategy is defined, identifying the real-time constraints to be respected in order to comply with the required functionality and general assumptions. The pre-architectures are generated based on templates, which allow the estimation of performance, power consumption, and circuit area for different design strategies. In this phase, the compiled code is mapped to hardware according to the minimum instruction set necessary to execute the application. The knowledge of architectural templates and target application provides useful information for dealing with design trade-offs.

In Phase 3 the profiling of the execution of the application code on the generated ASIPs is generated. Circuit characteristics are explored by interactive evaluation and refinement according to the data generated by hardware simulation. At this point, the designer can interact with the tool to explore different scenarios in terms of micro-architectures, building blocks, and custom ISA levels. This phase is repeated interactively after the modification of the hardware characteristics by the designer, such as the inclusion or removal of instructions in the ISA, the resizing of memory units, and the definition of data word length. The evaluation of execution latency for each operation provides an estimation of maximum operating frequency, as well as power consumption and circuit area. The design flow can be returned to Phase 1 if some possible code optimization is identified.

In Phase 4 the optimized hardware description of the ASIP is generated, which can be synthesized for FPGA or ASIC implementation, according to the application.

A customized compiler was developed specifically for the PAMPIUM microprocessor. The program described in C language is compiled to PAMPIUM assembly language and the variables are associated to registers according to the data type. At this point, a simulation of the assembly code provides some statistics about program behavior, such as the identification of the most frequent instruction types, the expected range of variables and the required number of registers. These results will guide the synthesis stage for architecture optimization.

Three different circuits are pre-generated according to the monocycle, pipeline and superscalar bases of PAMPIUM. A diagnosis of the specifications generated for these circuits provides information about the best organization that reach the design requirements. Finally, the hardware description of the ASIP is generated in SystemVerilog language [22], allowing the synthesis for both FPGA and ASIC implementation.

The developed compiler maps the variables according to their types to equivalent registers. The operations are also translated according to the target variable. Figure 16 shows the compilation flow implemented by the compiler.



Figure 16. Compilation flow of an input code performed by ASIPAMPIUM.

The compiler is capable to identify the number, type and scope of the variables. Some temporary variables in different scopes are allocated in order to share the same memory and register positions, thus providing a reduction in terms of required register bank and data memory sizes.

With the assembly simulator tool integrated in the ASIPAMPIUM it is possible to estimate the circuit performance for different hardware configurations, including circuit area, critical path, execution time, and static power consumption. Moreover, the simulator provides information about the minimal instruction set that must be implemented, the frequency of execution of a given instruction, the number of necessary registers, the frequency of memory accesses, the minimum and maximum values of the variables and the number and types of literal inputs in the application. These data provided by the simulator gives to the user a powerful tool for analyzing the trade-offs for circuit implementation.

For each microprocessor version an execution map is generated. It determines the number of executed instructions and the execution order. For the monocycle microprocessor, the execution map provides directly the number of program executions. For the pipeline version, the number of bubbles must be estimated, since it results in additional cycles for the execution of the same program. Pipeline bubbles are consequence of conditional jumps, which deviates the normal program execution flow, or data dependency. A form to reduce the number of bubbles is analyzing the expected result. The most frequent result for the conditional jump deviation test (true or false) is estimated in the simulation and used as standard for predicting if the deviation will occur or not. In the superscalar version the simulator performs an analysis of the parallel execution in the execution map, including bubble insertion analysis. It is possible to execute up to four instructions in parallel in this version. However, it must follow some rules related to data dependency. The superscalar version demands less cycles to execute a program, but it does not mean economy in terms of number of instructions. The execution time can be estimated by multiplying the delay of the critical path to the number of cycles necessary to execute a program.

A friendly graphical user interface containing the simulation results and trending graphs is available in the ASIPAMPIUM, helping the user to interpret data and take design decisions, as shown in Figure 17.

ASIPAMPIUM															3
≝ ⊒ ⊨ ▶ ↓ ५ ॥ ८ ⊤ ⊨															_
PAMPIUM Simulat	or a	and O	otimize	r i		Stop address		Current addr	ess	# of executions	;	# of literal in	puts		5
						505		506		7494319		Char Int	Float	Doub	ole
C Code				1								137895	5 315267	0	
as[5]=Asin[OF7];	ז   ו	– Assen	nply			Variables-					1	•			
OF8=Bi[7];		Lin	e Label	Assembly			Save	Load				Instructions			_
ac[7]=Acos[OF8];		494		ICOPY \$124(3),0;	•	Name	Address	Value	Max value	Min value		Instructions	# Lices	# True	
as[6]=Asin[OF8];		495		ICOPY \$124(2),0;		Temp_1	F\$XT1	53.827	594.39	-13.684 6 🔺		NOP	1	0	
		496		ICOPY \$124(1),0;		Temp_1	I\$XT1	14	31080	0 1:		CALL	56345	0	
return(larg);		497		ICOPY \$124(0),0;		Temp_2	I\$XT2	151	159	0		ECOPY	130194	0	
}		498		FCOPY \$F26,0;		Temp_2	F\$XT2	1.6689	12.691	-13.379 1		ICOPY	147771	0	
		499		FCOPY \$F26,\$I10;		Temp_3	F\$XT3	-0.002135	0.39101	-0.33207		RET	56344	0	
float findAB(int AP1){		500		FBSE \$F26,180,END_		rd	SI1	5918	31693	-1474		IBNE	31	0	
int AB=0;		501		FSUB \$F26,\$F26,360;		11	\$12	116	117	0		FBSF	102033	94209	
AP1=AP1/5:		502		FMULT \$F26,0.828571	1	12	\$13	101	101	0		FMULT	3376854	0	
AP1=AP1*5;		503		JUMP END_ELSE_25	;	13	\$14	122	124	0		IUMP	371305	0	
switch(AP1)		504	END_IF.	FMULT \$F26,0.828571	1 =	14	\$15	159	159	0		ISUB	16074	0	
case 0:		505	END_E.	. FMULT \$F25,0.017453	3	15	\$16	256	256	0		FBS	915	0	Ξ
AB = 14:		# 506		ICOPY \$123,1;		16	\$17	212	212	0		FBLE	3864	3618	
break		507		JUMP INI FOR 7;		17	\$18	110	126	0			651042	0	
case 5:		508	FOR_7/	IADD \$123,\$123,1;		18	\$19	169	169	0		IBLE	32023	7507	
AB = 16		509	_	IBLE \$123,120,END F	c	BETA	\$I10	0	0	0			7367	0	
break:		510	INI_FO	ISUB I\$XT1,\$123,1;		OF2	\$I11	0	0	0		MUT	425548	0	
case 10		511	_	FBS \$F1(I\$XT1),\$F25,	E	OF3	\$I12	0	0	0		FADD	1276223	0	
AB = 16		512		FBLE \$F1(\$I23),\$F25,	E	OF4	<b>\$I13</b>	0	0	0		IRI	246611	40326	
break:		513		ISUB \$122,\$123,1;		OF5	\$I14	0	0	0			10	43320	
case 15:		514		JUMP END FOR 7;		OF6	\$I15	0	0	0			10/1921	0	_
AB = 16		515	END IF.	JUMP FOR 7;		OF7	\$I16	0	0	0			2/0200	0	
break:		516	END F.	FCOPY \$F25.\$F1(\$122	2	OF8	\$I17	0	0	0			19640	24640	
case 20		517	_	ISUB \$118,\$122,60;		AP	\$I18	90	180	- 0		1D3E	40040	24040	
AB = 20		518		IBLE \$118,0,END IF 2		•		III		4					=
break:		519		IMULT \$128,-1,\$118:		Execution					_				
race 25-		520		CALL findAB;	-	Wat	ting								
		•		•											

Figure 17. ASIPAMPIUM graphical user interface.

## Design Strategies

The customized processor generated by ASIPAMPIUM tool is dependent on the quality of the input user application described in C language. In order to exemplify different optimization strategies, we present an example of the implementation of the cosine function using Taylor series.

The Taylor series that corresponds to the cosine function is given by:

$$\cos x = \sum_{n=0}^{\infty} \frac{-1^n}{(2n)!} x^{2n}; \tag{1}$$

A C code to calculate this series is shown in Algorithm 1.

Algorithm 1: Cosine function using Taylor series

The C code is implemented with a loop of type "for" and two subroutines, *fat2* and *exp2*. Six iterations were defined for the sum in order to obtain an error below  $10^{-6}$  [23]. Algorithm 2 presents the PAMPIUM assembly code generated by the compilation, with a total of 52 operations.

Algorithm 2: PAMPIUM assembly code for Algorithm 1
--

1	NOD
2	ICODY \$11.6. // main
2	
3	
4	COPT \$F1,0.785395163397;
5	
6	COPY \$12,0;
7	JUMP INL_FOR_1;
8	IADD \$12,\$12,1; // FOR_1
9	IBL \$12,6,END_FOR_1;
10	IREST 1\$X11,\$12,2; // INI_FOR_1
11	IBNE ISXT1,0,END_IF_1;
12	ICOPY \$13,\$12;
13	CALL fat2;
14	
15	ICOPY \$13, \$12;
16	RCOPY \$F3, \$F1;
17	CALL Exp2;
18	FDIV F\$X11, \$X11, 1\$X11;
19	FADD \$F2, F\$XT1, \$F2;
20	JUMP END_ELSE_1;
21	COPY \$13,\$12; // END_IF_1
22	CALL fat2;
23	
24	
25	
26	CALL EXP2;
2/	
20	$FSUD \Rightarrow FZ, \Rightarrow FZ, FZ, FSAT1;$ IF MD FOR 1. (J FAIL) FILE 1.
29	JUNIF FOR_1; // END_ELSE_1
21	ENU; // ENU_FON_1
22	INDEL 514 (2,515) // SUD-INTERIOR TAL2
22	
24	
35	IMP INFOR 2.
36	ADD SIG SIG 1: // FOR 2
37	IBL SIG SIG FUID FOR 2.
38	MULT \$15 \$16 \$15 \(/ INI FOR 2
39	IMP FOR 2:
40	ICOPY_SATL_SI5: // END_FOR_2
41	RET:
42	MULT \$F4.\$F3.\$F3: // sub-function Exp2
43	FCOPY \$F5.1:
44	ICOPY \$14.0;
45	ICOPY \$14.1:
46	UMP INI FOR 3;
47	IADD \$14, \$14, 1; // FOR_3
48	IBL \$14,\$13,END_FOR_3;
49	FMULT \$F5,\$F4,\$F5; // INI_FOR_3
50	JUMP FOR_3;
51	PCOPY \$XT1, \$F5; // END_FOR_3
52	RET;

The customized ASIP to implement the cosine function presents a 32-bit fixed and floating point architecture. It is necessary the use of 12 32-bit registers and 5 8-bit registers. The customized instruction set is composed of 20 instructions. The results generated by the ASIPAMPIUM tool for the three PAMPIUM versions are presented in columns 2, 3 and 4 of Table 4. It is possible to note that the monocycle version presents smaller area and smaller power consumption in comparison with other versions. However, it presents a slower execution time. As expected, the pipeline version presents intermediate values and the superscalar version has the smaller execution time, but at the expenses of increase in area and power consumption.

**Table 4.** Results for customized ASIP versions generated by ASIPAMPIUM for the cosine function using Taylor series.

Davamatav	Origiı	nal Code (Algor	ithm 1)	Optimized Code (Algorithm 3)				
rarameter	Monocycle	Pipeline	Superscalar	Multicycle	Pipeline	Superscalar		
Area	0.19 mm <sup>2</sup>	0.22 mm <sup>2</sup>	0.28 mm <sup>2</sup>	0.11 mm <sup>2</sup>	0.14 mm <sup>2</sup>	0.20 mm <sup>2</sup>		
Execution time	41.0 μs	29.0 µs	17.0 μs	13.0 µs	8.3 µs	7.2 μs		
Power consumption	5.2 mW	7.7 mW	11.0 mW	1.9 mW	3.3 mW	4.5 mW		

Figure 18a presents the total time execution for instruction type. Most of the execution time is spent with general-purpose instructions (literal load, *COPY*, etc.) and jump instructions (*JUMP*, *CALL*, *BLE* and *BL*), representing about 67% of the total time. Arithmetic instructions, such as *FADD*, *FDIV* and *ITOF* represent only 8% of the execution time. So, it is possible to notice that the elimination of loops and subroutines in the C code is mandatory to optimize the total execution time.

Figure 18b depicts the contribution of instruction types over the total microprocessor area. About 43% of the circuit is used by the floating point monocycle division block. The second larger block is the fixed point multiplier, occupying 24% of the circuit area. An optimization in the area could be performed by substituting the divider by its multicycle version—which presents smaller area—or by eliminating any floating point division in the code.

Figure 18c shows the contribution of each instruction type to the circuit power consumption. The block with larger energetic impact is again the monocycle floating point divider, with a total of 6  $\mu$ J. This value is very large compared to the remaining blocks, indicating that it can be substituted by the multicycle version, which presents better energetic efficiency.

Figure 18d presents the relative energy consumption for each instruction type. The floating point division is responsible for 39% of the total consumed energy, followed by the fixed point multiplication, with 31%, and the general purpose instructions, with 19%.

The monocycle floating point divider has the larger critical path (about 51 ns), as can be seen in Figure 18e. The remaining blocks present a critical path around 10 ns, which indicates a balance problem in relation to the instruction delays. It suggests a multicycle implementation for the divider, which allows the increasing in the microprocessor operation frequency.

So, rewriting the application code it is possible to improve the hardware performance in different aspects. The floating point division is used only few times in the algorithm, but presents a high impact on power consumption, area and delay. However, it is not possible to eliminate it, because it is fundamental for the Taylor series calculation. So, the best approach is the substitution of the monocycle divider block by its multicycle version.

Furthermore, most of the execution time is spent with loops and subroutines. The code can be optimized by removing unnecessary loops and by reorganizing functions for calculating exponent and factorial—which can take advantage of previous results for calculation.





Another important aspect which contributes to the circuit performance is the variable type for each operation. Presented results use 12 32-bit registers and the fixed-point numbers are also 32 bits. The factorial calculation generates very high numbers, which requires an increasing in area to represent the operands. As the fixed-point hardware is sized by the largest operand, this function would impact in unnecessary effort for the remaining functions. So, the best option is to change this variable to floating point. Algorithm 3 presents the refactored code with the improvements discussed before.

Algorithm 3: C	<b>Optimized</b>	cosine	function	for	hardware s	synthesis.
0						

1	float xfat=2;
2	float xt1=3;
3	void main(){
4	int n=6;
5	int i=0;
6	float x=0.6168502751;// (pi/4)^2;
7	float xt=x;
8	float res= $1-xt + 0.5$ ;
9	for $(i = 2; i \le n; i + +)$
10	xfat=xfat*xt1;
11	xt1=xt1+1;
12	xfat=xfat * xt1 ;
13	xt=xt*x;
14	if ((i%2)==0)
15	res=res+xt/xfat;
16	else
17	res=res-xt/xfat;
18	}
19	

Columns 5, 6 and 7 of Table 4 summarizes the new circuit performance, considering code optimization. The values of area, execution time and power consumption were significantly reduced compared to the non-optimized code. Multicycle version presents the smaller area and power consumption (reduction of 42% and 36%, respectively, in comparison to the original monocycle version), while superscalar version presents the best performance in terms of execution time, with a reduction of more than 2x in relation to the original counterpart. This demonstrates that the statistical analysis generated by ASI-PAMPIUM tool provides very useful information for the optimization of customized ASIPs.

Design strategies related to C code optimization can be generalized for any application. The elimination of a given arithmetic operation or its substitution by the multicycle version, when possible, can result in great improvements in terms of area and power consumption. The removal of unnecessary loops contributes to the reduction of jump addresses calculation, thus providing advantages in terms of power consumption and performance. The reuse of temporary variables reduces the size of the register bank. Furthermore, the type of numerical variables (integer or floating point) must be evaluated in relation to area, power consumption and performance. However, the application of these strategies depends on design objectives. A careful evaluation of the resulting effect over conflicting requirements is mandatory.

## 5. Application Example: FFT Synthesis

This section presents the results generated by the ASIPAMPIUM tool for the implementation of a dedicated Fast Fourier Transform (FFT).

The FFT is applied to several data transmission systems. It transforms the data in the time domain to the frequency domain [24]. This technique is very used for the analysis of the modulation signal and for orthogonal frequency division multiplexing (OFDM), which is the base for different communication systems [25]. Several digital communication standards, such as ultra-wideband (UWB), worldwide interoperability for microwave access (WiMAX), wireless LAN (WLAN), integrated services digital broadcasting terrestrial (ISDB-T), long-term evolution (LTE), digital video broadcasting (DVB-T), and digital video broadcasting second generation terrestrial (DVB-T2) adopt OFDM systems for the processing of digital signals [26].

The discrete-time Fourier transform Z(k) of a finite number of points X(n), where  $n \in [0, N-1]$ , is calculated as:

$$Z(k) = \sum_{n=0}^{N-1} X(n) W_N^{nk} \quad (k \in [0, N-1])$$
<sup>(2)</sup>

The coefficient  $W_N^{nk}$  is a complex number represented by the following exponential equation:

$$W_N^{nk} = e^{\frac{-2j\pi nk}{N}} \tag{3}$$

In general, the complexity of FFT evaluation can be reduced to two sums of *M* and *L* coefficients, where the number of points *N* is given by  $N = M \cdot L$ . So,

$$Z(k) = Z(s + M \cdot p)$$

$$= \sum_{l=0}^{L-1} \sum_{m=0}^{M-1} x(l + L \cdot m) W_{M \cdot L}^{(l+L \cdot m)(s+M \cdot p)}$$

$$= \sum_{l=0}^{L-1} \sum_{m=0}^{M-1} x(l + L \cdot m) W_{N}^{s \cdot l} W_{M}^{s \cdot m} W_{L}^{l \cdot p}$$

$$= \sum_{l=0}^{L-1} W_{N}^{s \cdot l} [\sum_{m=0}^{M-1} x(l + L \cdot m) W_{M}^{s \cdot m}] W_{L}^{l \cdot p}$$
(4)

where  $N = L \cdot M$ ,  $k = s + M \cdot p$ ,  $n = l + L \cdot m$ .

A radix-16 algorithm was implemented, using variables of type *float* for the calculation of the FFT points and variables of type *int* for the calculation of the FFT indexes.

The algorithm is able to calculate an FFT with N points, in which N is a power of 2 from 32 to 2048. A 16-point butterfly was optimized in order to reduce the number of operations and variables.

With this algorithm, it is necessary to execute 1024 accesses to data memory and 512 floating point multiplications and sums for each butterfly, not considering the calculation of the indexes.

In order to reduce even more the computational effort, the calculation is divided into real and imaginary parts, where  $Z(k) = X(k) + j \cdot Y(k)$ . With this simplification, it is possible to reuse some twiddle factors. The number of data memory accesses per butterfly is reduced to only 16 (the size of the butterfly), the number of floating point multiplications to 48 (since some twiddle factors are zero or repeated), and the number of sums to 368, thus minimizing the computational effort to calculate the FFT of a discrete signal.

For example, for a complete calculation of a 2048-points FFT, it is necessary  $16 \cdot 64 \cdot 2048 = 2^{21}$  memory accesses using the described methodology. Otherwise, without this technique, it would be necessary  $1024 \cdot 64 \cdot 2048 = 2^{27}$  memory accesses. Thus, the number of memory accesses is reduced by 64 times.

The developed program has about 300 lines of C code, and its translation to Assembly resulted in 3071 operations. Only 23 different instructions were necessary to implement the code, from the 80 available in the PAMPIUM architecture. The most frequent instruction is the floating point add (FADD). The simulations and analyses performed by the ASIPAMPIUM tool last about 3 h in an i7-3770 CPU @ 3.40 GHz.

The resulting processor synthesized by the ASIPAMPIUM tool has 9 16-bit integer registers, 81 32-bit floating point registers, and 6 8-bit registers.

Figure 19a presents the relative area occupied by each logic block. The register bank block is the most area consuming, due to the high number of registers to store the program variables. So, optimization in terms of circuit area could be obtained by reducing the number of variables in the code.

Figure 19b presents the execution time spent for each instruction type. General type instructions (literal input and data memory access) require about 47% of the total execution time. The most used arithmetic instruction is the floating point add, demanding 17% of the execution time. An optimization in the execution time could be obtained by reducing the number of data memory access or the literal inputs.

Figure 19c presents the energy by operation consumed by the implemented logic blocks, showing that integer and floating point multiplication are the most energy-consuming instructions.

The relative energy consumption of each logic block in relation to the overall microprocessor consumption is presented in Figure 19d. The general purpose instructions are responsible for most of the energy consumed in an FFT calculation, demanding 36% of the total energy.

In terms of delay, the floating point add is the slower block, as depicted in Figure 19e. It can be seen that some optimizations can be achieved for the FFT calculation in the code domain, mainly to reduce the number of arithmetical operations.

The identification of the common variables in the algorithm and the reuse of values generated by some arithmetical operations are key for reducing the time- and area-consuming instructions. So, the number of floating point sums and multiplications inside the butterfly can be reduced to 94 and 18, respectively, as well as the number of memory accesses can be minimized to 16.

After optimizing the code, we generated two versions of the PAMPIUM processor: a multicycle version, called **PAMPIUM\_M**, optimized in terms of circuit area and power consumption; and a superscalar version, called **PAMPIUM\_S**, developed to achieve the best possible performance.

Area (%)

(Ln)

Energy per operation



Instruction type (d)



INULI IADD

(c)

IMULT

(a)



In order to compare the obtained results with other works described in the literature, we adopt as a figure of merit the normalized energy  $\hat{E}$  and area  $\hat{A}$  [27]. Both are referred to 65 nm technology node and take into account the voltage scaling:

$$\widehat{E} = \frac{P \times T_{clock} \times N_{FFT}}{\frac{Tech.Node}{65nm} \times \left(\frac{V_{DD}}{0.9}\right)^2}$$
(5)

$$\widehat{A} = \frac{A}{\left(\frac{Tech.Node}{65nm}\right)^2} \tag{6}$$

Here, P is the power consumption,  $T_{clock}$  is the clock period,  $N_{FFT}$  is the number of clock cycles necessary to calculate the FFT,  $V_{DD}$  is the voltage supply and A is the circuit area.

2

The obtained results are compared to the work in [28], which describes the implementation of a module to perform FFT calculation using the CORDIC (Coordinate Rotation *Digital Computer*) method using Xilinx Virtex-5 FPGA platform.

Table 5 presents the results and the comparison with the versions generated by ASIPAMPIUM, which are implemented in Altera Stratix IV FPGA. These versions are

based on multicycle (**PAMPIUM\_M**) and superscalar (**PAMPIUM\_S**) architectures. The **PAMPIUM\_M** version presents a smaller area with respect to [28] (smaller number of lookup tables—LUTs), but at the expense of smaller operation frequency. The **PAMPIUM\_S** version, on the other side, is faster but demands a higher number of LUTs. It can be seen that there is a trade-off between area and performance in the FPGA implementation.

Table 5. FPGA implementation results for the FFT algorithm.

Parameter	PAMPIUM_M	PAMPIUM_S	[28]
Points	32-2048	32-2048	64-8092
LUTs	9125	14,125	13,176
Registers	4033	4033	3486
Operating frequency	95 MHz	135 MHz	131 MHz

It is also possible to compare the results obtained by the physical synthesis of the **PAMPIUM\_M** and **PAMPIUM\_S** in 180 nm fabrication technology with other state-of-theart works in the literature. The work described in [13] presents a hardware description language of the LISA machine for the implementation of an ASIP. Using software tools for design, system verification, and implementation, the proposed design flow allows prototyping and testing, aiming to reduce design time and human effort with respect to traditional design methodologies. It presents a case study on a parallel FFT algorithm. In [26] a reconfigurable process platform is used and a similarity matrix is adopted with the goal to reduce the number of access to the main memory by using a large register file to store intermediate data. In [29] a radix-4/2 butterfly structure is adopted for exploring the parallelism in the FFT calculation, but it requires more memory accesses. In [30], four VLIW vias and four tracks in the SIMD architecture are adopted, with different data shuffling modes to improve the FFT transfer rate, aiming to reduce energy consumption.

Table 6 summarizes the comparison between these implementations and the generated PAMPIUM versions using the normalization Equations (5) and (6).

Parameter	PAMPIUM_M <sup>1</sup>	PAMPIUM_S <sup>2</sup>	[13]	[26]	[29]	[30]
Tech. node	180 nm	180 nm	65 nm	130 nm	130 nm	90 nm
Points	32–2048	32–2048	16–2048	16–4096	16–4096	64–2048
Oper. freq. (MHz)	55	98	470	320	100	350
Power (mW)	8.8	19.5	51.0	60.7	87.2	106.5
$N_{FFT}$ (1k FFT)	33,600	14,280	1024	4526	1280	1950
$\widehat{E}$ (nJ)	175.3	92.6	111.1	120.7	157.0	174.1
Area (mm <sup>2</sup> ) $\widehat{A}$	0.19 0.03	0.34 0.04	0.75 0.75	_	2.23 0.56	2.45 1.28

Table 6. Physical synthesis results for the ASIP implementation of the FFT algorithm.

<sup>1</sup> PAMPIUM multicycle, <sup>2</sup> PAMPIUM superscalar.

We can notice that the multicycle version of PAMPIUM, synthesized for 180 nm technology, presents the smallest circuit area. The operation frequency is lower, due to the smaller parallelism level in comparison to the other implementations. However, the extremely low power consumption of 8.8 mW indicates the optimization for low-power applications while occupying the smaller silicon area. This version needs 33,600 cycles ( $N_{FFT}$ ) to perform the calculation of a 1024-point FFT, which degrades the energy efficiency.

The superscalar version of PAMPIUM, on the other side, presents better energy efficiency compared to the other implementations, while maintaining a small silicon area. It has a power consumption of just 19.5 mW at an operating frequency of 98 MHz. The instruction-level parallelism reduces the  $N_{FFT}$  to less than half of the multicycle version.

The good results obtained by the ASIPAMPIUM for the generated ASIPs for implementing the FFT algorithm demonstrate the functionality of the EDA tool. It provides an efficient environment for dealing with design trade-offs in a short design time, generating optimized circuits with a performance comparable to the obtained by other methodologies.

## 6. Prototyping and Measurement Results

We prototyped a version of PAMPIUM generated by ASIPAMPIUM in order to validate in silicon the complete design methodology implemented with the tool. The central application is presented in Figure 20, which is a hardware platform for gesture recognition system using accelerometers and gyroscopes for the detection of movements in the *x*, *y*, and *z* axes. The system must read up to six sensors using I2C protocol and transmit the information to a computer through the RS232 serial protocol with a power budget of 50 mW.



Figure 20. Sensor system using PAMPIUM.

The version of the processor, called PAMPIUM IC, was developed using the ASI-PAMPIUM tool and the circuit was prototyped in TSMC 0.18  $\mu$ m technology. To provide the reading of the sensors, three interface modules were implemented, compatible with I2C protocol [31].

The entire circuit has a hard area constraint of 0.34 mm<sup>2</sup>, as well as a limit of 20 input/output pins.

Small cache memories are implemented inside the processor, reducing the silicon area. They communicate with data memory and program memory, which are located outside the chip, by means of the SPI protocol.

Logic and physical synthesis of the circuit, as well as simulation, were executed using Synopsys RTL-to-GDSII flow, with the digital logic library provided by the foundry.

A C program was developed for the implementation of three I2C and one RS232 communication module.

The processor architecture is of integer type, with a data length of 16 bits and an instruction word of 24 bits. It has 64 16-bit registers, 9 of them used specifically for literal input, program memory pointer, and configuration. Figure 21 presents the implemented register bank.

Registers *CONFIG\_0* and *CONFIG\_1* are reserved for the implementation of configuration signals and interruption control. Registers *DATA\_I2C\_0*, *DATA\_I2C\_1*, *DATA\_I2C\_2* and *DATA\_RS232* are reserved for the data exchange between the processor and the communication modules. Register *PORT\_IN/OUT* is used for the configuration of directions and values of the input and output pins. Registers *REG\_LITERAL* and *POINTER* have the function to store the input literal and the data memory pointer, respectively.

The RTL description was divided into two stages. First, the RTL description of the monocycle version of PAMPIUM was generated by the ASIPAMPIUM tool. The processor was optimized for achieving the area constraint. In the second stage, the cache memory was included, as well as the blocks for the interface with RS232 and I2C protocols.

0	CONFIG_0
1	CONFIG_1
2	DATA_I2C_0
3	DATA_I2C_1
4	DATA_I2C_2
5	DATA_RS232
6	PORT_IN/OUT
•	•
•	•
•	•
•	•
62	REG_LITERAL
63	POINTER

Figure 21. Register bank implemented for the PAMPIUM IC.

The data cache memory is composed of 16 positions, each one storing a 16-bit word. These positions are divided into four banks. Each bank has a register that store the upper part of the data memory address. So, the accesses performed by the external memory are made in blocks of four consecutive accesses. This strategy maximizes the performance, exploring the local proximity of the data.

The program cache memory is composed of eight positions, each one containing a 24-bit word. Each position has a register for storing its original address. The storing process is made in a sequential form, i.e., the instructions are stored from the lower to the higher address, due to the sequential nature of the program memory. The use of eight store positions allows the execution of small loops without the need to perform additional external loads.

The implementation of the data and program cache memories, with the remaining memory in the external part of the chip communicating by means of SPI protocol, allows the use of only eight input/output pins for data transfer between memory and CPU. With the remaining pins of "clk", "rst", "vdd" and "gnd", and the four communication modules, the overall circuit needs a total of 20 input/output pins. Figure 22 presents the final pin scheme of the customized processor.



Figure 22. PAMPIUM IC input and output pins.

 Image: set of the set of

The customized ISA is composed of 21 instructions. The layout of the PAMPIUM IC obtained after the physical synthesis for 0.18  $\mu$ m technology is presented in Figure 23a.

(c)

**Figure 23.** PAMPIUM IC layout and chip implementation. (**a**) Layout of the PAMPIUM IC. (**b**) Complete layout of the multi-user chip. (**c**) Photography of the prototyped chip. (**d**) Test setup PCB.

(d)

Three clock trees were considered: program memory clock (Flash type), data memory clock (SRAM), and internal clock. Maximum delays of 10 ns, 20 ns, and 100 ns were determined for program memory, data memory, and internal clock, respectively.

The minimum clock frequency required on the physical synthesis was defined as 80 MHz. This constraint is necessary to deal with the external memory operation frequencies. The internal clock divider is configured as follows: the main memory clock is equal to the main clock, the data memory clock is two times smaller than the main clock, and the processor clock is eight times slower than the main clock. So, the cache module for data memory is fed with a clock of 40 MHz and the internal processor clock works at 10 MHz.

The results obtained by the physical synthesis are summarized in Table 7. The maximum clock frequency achieved 102.5 MHz, with an overall power consumption of 19 mW. It meets the performance requirement. The final layout achieved an area of 500  $\mu$ m × 627  $\mu$ m = 0.314 mm<sup>2</sup>, which is inside the area constraint.

The circuit was validated for the timing constraints and the communication modules were tested, as well as the instruction behavior. The final layout of the prototyped chip is shown in Figure 23b. The PAMPIUM IC is located in the right inferior part of the chip.

The chip was prototyped by the mini@sic multi-user program from IMEC. Figure 23c shows the microphotography of the final chip and Figure 23d shows the test setup PCB used to extract measurement results. The measured power consumption is 28 mW for a voltage supply of 1.8 V and an operating frequency of 80 MHz. It is slightly higher than the value estimated by simulation but still meets the design power budget.

Table 7. Physical synthesis results for PAMPIUM IC.

Result	Value
Area	0.314 mm <sup>2</sup>
Maximum operating frequency	102.5 MHz
Mean power consumption	19 mW

### 7. Conclusions

This work described the development of ASIPAMPIUM, a fast ASIP generator tool, which is suitable for the design of optimized dedicated processors. The tool provides a series of features, such as compilation, simulation, and synthesis of a C code to logic level. The generated ASIP is based on PAMPIUM architecture, which can be customized for different applications. The processor optimization in terms of low power consumption and small silicon area can be achieved by exploring instruction-level parallelism, customized ISA, and data word length.

The application of ASIPAMPIUM for generating a customized FFT ASIP shows the capabilities of the tool. The superscalar version presented the lowest normalized energy consumption compared to other implementations described in the literature. Moreover, the multicycle version presented the smallest area for the calculation of a 2048-point FFT.

The prototyping of a version of PAMPIUM in silicon demonstrated the complete design flow, from the C code program to the physical implementation. It implements a dedicated ASIP with three I2C interfaces for reading six sensors in real-time, a SPI interface for transferring data to memory, and an RS232 interface for communicating with an external device. The whole system consumes 28 mW for an operating frequency of 80 MHz.

The proposed tool, in combination with the proposed ASIP design methodology, contributes to the fast implementation of dedicated hardware suitable for mobile and IoT applications. It reduces the time-to-market and provides a simple and intuitive design flow, allowing the designer to handle different design trade-offs and objectives.

Since the current version of the ASIPAMPIUM tool is limited to single-core applications, future improvements must include the possibility to deal with adaptable ASIP-based MPSoC technologies, expanding the application range for ultra-high-performance integrated systems. Furthermore, the support for hardware accelerators will provide more alternatives to the designer in the synthesis of an optimized ASIP. The development of a custom backend for the PAMPIUM ISA for a traditional optimizing compiler (such as GCC) is also an important step toward the wide adoption of the proposed methodology.

Author Contributions: Conceptualization, A.E. and A.G.; methodology, A.G.; software, A.E.; validation, A.E. and M.R.; formal analysis, A.E.; investigation, A.G., L.C.-S. and P.C.C.d.A.; resources, A.G., L.C.-S. and P.C.C.d.A.; data curation, A.E.; writing—original draft preparation, A.E., M.R. and A.G.; writing—review and editing, L.C.-S. and P.C.C.d.A.; visualization, M.R.; supervision, A.G., L.C.-S. and P.C.C.d.A.; project administration, A.G.; funding acquisition, A.G. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by Brazilian Research Agencies CNPq (Conselho Nacional de Desenvolvimento Científico e Tecnológico) and FAPERGS (Fundação de Amparo à Pesquisa do Estado do Rio Grande do Sul), grant number 22/2551-0000841-0. The APC was funded by FAPERGS.

**Conflicts of Interest:** The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

# Abbreviations

The following abbreviations are used in this manuscript:

ALU	Arithmetic and Logic Unit
ASAM	Automatic Architecture Synthesis and Application Mapping
ASIC	Application Specific Integrated Circuit
ASIP	Application Specific Instruction Set Processor
CISC	Complex Instruction Set Computer
CORDIC	Coordinate Rotation Digital Computer
CPU	Central Processing Unit
CMOS	Complementary Metal-Oxide-Semiconductor
DVB-T	Digital Video Broadcasting Terrestrial
DVB-T2	Digital Video Broadcasting Second Generation Terrestrial
EDA	Electronic Design Automation
FFT	Fast Fourier Transform
FPGA	Field-Programmable Gate Array
FPU	Floating-Point Unit
HW	Hardware
ILP	Instruction Level Parallelism
IoT	Internet of Things
ISDB-T	Integrated Services Digital Broadcasting Terrestrial
ISA	Instruction Set Architecture
LAN	Local Area Network
LTE	Long Term Evolution
LUT	Look-Up Table
MIMD	Multiple Instruction Multiple Data
MGP	Multigrained Profiling
MPSoC	Multiprocessor System on Chip
OFDM	Orthogonal Frequency Division Multiplexing
RISC	Reduced Instruction Set Computer
RTL	Register Transfer Level
SIMD	Single Instruction Multiple Data
SOC	System-on-Chip
SW	Software
UWB	Ultra Wide Band
VLIW	Very Long Instruction Word
WLAN	Wireless Local Area Network
WiMAX	Worldwide Inter-operability for Microwave Access

### References

- Jordans, R.; Jóźwiak, L.; Corporaal, H.; Corvino, R. Automatic Instruction-Set Architecture Synthesis for VLIW Processor Cores in the ASAM Project. *Microprocess. Microsystems* 2017, *51*, 114–133. [CrossRef]
- Hamed, A.; El-Kharashi, M.W.; Salem, A.; Safar, M. Two-Layer Bus-Independent Instruction Set Architecture for Securing Long Protocol Data Units in Automotive Open System Architecture-Based Automotive Electronic Control Units. *Electronics* 2022, 11, 952. [CrossRef]
- 3. Mazo, J.C.; Leupers, R. *Programming Heterogeneous MPSoCs: Tool Flows to Close the Software Productivity Gap*; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2013.
- Karuri, K.; Leupers, R. Application Analysis Tools for ASIP Design: Application Profiling and Instruction-Set Customization; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2011.
- 5. Wolf, M. Computers as Components: Principles of Embedded Computing System Design; Elsevier: Amsterdam, The Netherlands, 2012.
- 6. Shahabuddin, S.; Mammela, A.; Juntti, M.; Silven, O. ASIP for 5G and Beyond: Opportunities and Vision. *IEEE Trans. Circuits Syst. II Express Briefs* **2021**, *68*, 851–857. [CrossRef]
- Eusse, J.F.; Williams, C.; Leupers, R. Coex: A Novel Profiling-based Algorithm/Architecture Co-Exploration for ASIP Design. ACM Trans. Reconfigurable Technol. Syst. (TRETS) 2015, 8, 17. [CrossRef]
- Jozwiak, L.; Lindwer, M.; Corvino, R.; Meloni, P.; Micconi, L.; Madsen, J.; Diken, E.; Gangadharan, D.; Jordans, R.; Pomata, S.; et al. ASAM: Automatic Architecture Synthesis and Application Mapping. *Microprocess. Microsystems* 2013, 37, 1002–1019. [CrossRef]
- 9. Jóźwiak, L.; Nedjah, N.; Figueroa, M. Modern Development Methods and Tools for Embedded Reconfigurable Systems: A Survey. *Integr. VLSI J.* **2010**, *43*, 1–33. [CrossRef]

- Eusse, J.F.; Fernandez, F.; Leupers, R.; Ascheid, G. Concurrent memory subsystem and application optimization for ASIP design. In Proceedings of the 2016 International Conference on Embedded Computer Systems: Architectures, Modeling and Simulation (SAMOS), Samos, Greece, 17–21 July 2016; pp. 1–10.
- 11. Muller, O.; Baghdadi, A.; Jezequel, M. From Parallelism Levels to a Multi-ASIP Architecture for Turbo Decoding. *Very Large Scale Integr. (VLSI) Syst. IEEE Trans.* 2009, 17, 92–102. [CrossRef]
- Wang, Y.; Ha, Y. A Performance and Area Efficient ASIP for Higher-Order DPA-Resistant AES. *Emerg. Sel. Top. Circuits Syst. IEEE J.* 2014, 4, 190–202. [CrossRef]
- Chen, T.; Pan, X.; Liu, H.; Wu, T. Rapid prototype and implementation of a high-throughput and flexible FFT ASIP based on LISA 2.0. In Proceedings of the IEEE 2014 15th International Symposium on Quality Electronic Design (ISQED), Santa Clara, CA, USA, 3–5 March 2014; pp. 681–687.
- Pees, S.; Hoffmann, V.; Zivojnovic, V.; Meyr, H. LISA—Machine description language for cycle-accurate models of programmable DSP architectures. In Proceedings of the 36th Anual ACM/IEEE Design Automation Conference, DAC 99, New Orleans, LO, USA, 21–25 June 1999; pp. 933–938.
- 15. Davidson, S.; Batten, C.; Taylor, M.B. The Celerity Open-Source 511-Core RISC-V Tiered Accelerator Fabric: Fast architectures and design methodologies for fast chips. *IEEE Micro* 2018, *38*, 30–41. [CrossRef]
- Asanović, K.; Avizienis, R.; Bachrach, J.; Beamer, S.; Biancolin, D.; Celio, C.; Cook, H.; Dabbelt, D.; Hauser, J.; Izraelevitz, A.; et al. *The Rocket Chip Generator*; Technical Report UCB/EECS-2016-17; EECS Department, University of California: Berkeley, CA, USA, 2016.
- Hepola, K.; Multanen, J.; Jaaskelainen, P. OpenASIP 2.0: Co-Design Toolset for RISC-V Application-Specific Instruction-Set Processors. In Proceedings of the International Conference on Application-Specific Systems, Architectures and Processors, Gothenburg, Sweden, 12–14 July 2022; pp. 161–165. [CrossRef]
- 18. Synopsys. ASIP Designer: Application-Specific Processor Design Made Easy. 2022. Available online: https://www.synopsys. com/designware-ip/processor-solutions/asips-tools.html (accessed on 8 December 2022).
- Codasip. RISC-V Processor Customization. 2022. Available online: https://codasip.com/solutions/processor-customization/ (accessed on 8 December 2022).
- Andes Technology. Custom Computing. 2022. Available online: http://www.andestech.com/en/products-solutions/customcomputing/ (accessed on 8 December 2022).
- 21. Engroff, A.M.; Girardi, A.G.; Heckler, M.V.; Winterstein, A.; Greda, L.A. ASIP development of a real-time control module for a retrodirective antenna array. *AEU—Int. J. Electron. Commun.* **2019**, 109, 31–42. [CrossRef]
- 22. Sutherland, S.; Moorby, P.; Davidmann, S.; Flake, P. SystemVerilog for Design Second Edition: A Guide to Using SystemVerilog for Hardware Design and Modeling; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2006.
- 23. Hazewinkel, M. Encyclopedia of Mathematics; Springer: Berlin/Heidelberg, Germany, 2001.
- 24. Walker, J.S. Fast Fourier Transforms; CRC Press: Boca Raton, FL, USA, 1996.
- 25. Son, B.S.; Jo, B.G.; Sunwoo, M.H.; Kim, Y.S. A high-speed FFT processor for OFDM systems. In Proceedings of the 2002 IEEE International Symposium on Circuits and Systems (ISCAS), Phoenix-Scottsdale, AZ, USA, 26–29 May 2002; Volume 3, p. III.
- 26. Guan, X.; Fei, Y.; Lin, H. Hierarchical design of an application-specific instruction set processor for high-throughput and scalable FFT processing. *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.* **2012**, *20*, 551–563. [CrossRef]
- 27. Chen, Y.; Lin, Y.W.; Tsao, Y.C.; Lee, C.Y. A 2.4-Gsample/s DVFS FFT processor for MIMO OFDM communication systems. *IEEE J. Solid-State Circuits* 2008, 43, 1260–1273. [CrossRef]
- Chen, J.; Lei, Y.; Peng, Y.; He, T.; Deng, Z. Configurable Floating-Point FFT Accelerator on FPGA Based Multiple-Rotation CORDIC. *Chin. J. Electron.* 2016, 25, 1063–1070. [CrossRef]
- Bo, Y.; Han, J.; Zou, Y.; Zeng, X. A low power ASIP for precision configurable FFT processing. In Proceedings of the 2012 Asia-Pacific Signal & Information Processing Association Annual Summit and Conference (APSIPA ASC), Hollywood, CA, USA, 3–6 December 2012; pp. 1–4.
- Zhang, B.; Liu, H.; Zhao, H.; Mo, F.; Chen, T. Domain specific architecture for next generation wireless communication. In Proceedings of the 2010 Design, Automation & Test in Europe Conference & Exhibition (DATE 2010), Dresden, Germany, 8–12 March 2010; pp. 1414–1419.
- 31. Null, L.; Lobur, J. The Essentials of Computer Organization and Architecture; Jones & Bartlett Publishers: Burlington, MA, USA, 2014.

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.